Author's response to reviews

Title: "Done more for me in a fortnight than anybody done in all me life." How welfare rights advice can help people cope with cancer.

Authors:

Suzanne Moffatt (s.m.moffatt@ncl.ac.uk)
Emma Noble (e.j.noble@ncl.ac.uk)
Catherine Exley (catherine.exley@ncl.ac.uk)

Version: 4 Date: 25 August 2010

Author's response to reviews:

Dear Editor

Thank you for accepting our paper.

All requested changes have been made to the manuscript as follows:

Major revision: provide a functioning web link to the website for Reference 23 - done

Minor revisions: format abbreviations into a paragraph - done

Tables - remove vertical lines from tables - done

Additional files removed - done

The paper has been thoroughly proof read and typographical corrections made. There is one copy of the paper, filename, 'MoffattetalAugust2010FINALREVISIONPROOFS'

We look forward to seeing the paper in print.

Thank you for all your help and the efficiency with which you have dealt with our submission.

Best wishes

Suzanne Moffatt

Emma Noble

Catherine Exley